Author's response to reviews

Title: Sexual health for women with Rheumatoid Arthritis in relation to physiotherapy - a qualitative study

Authors:

Kristina I Areskoug-Josefsson (kristina.areskoug@telia.com)
Gunvor Gard (gunvor.gard@med.lu.se)

Version: 5 Date: 7 October 2010

Author's response to reviews: see over
Cover letter

Dear Biomed Central Editorial

We thank You for all valuable comments on our manuscript and all suggestions for improvements. We have now improved the manuscript according to Your suggestions and hope that it now can be published.

The following improvements have been performed:

Title, author list and abstract are identical in the manuscript and the submission system.

Highlighting/tracking is removed from the manuscript.

References are corrected.

Table 1 is removed and incorporated in the text.

Contributions, renamed to 'Authors' contributions'.